Author's response to reviews

Title: Anteroposterior glide versus rotating platform low contact stress (LCS) knee arthroplasty: a randomised controlled trial

Authors:

Sandeep Datir (datirsandeep@hotmail.com)
Gayle Walley (g.walley@keele.ac.uk)
Murali Krishna Sayana (sayanamk@gmail.com)
Aziz Rahmatalla (a.rahmatalla@orth.keele.ac.uk)
Ian DosRemedios (dosremedios@ntlworld.com)
Charles Wynn-Jones (chwj@doctors.org.uk)
Stephen Bridgman (stephen.bridgman@northstaffs.nhs.uk)
Nicola Maffulli (osa14@keele.ac.uk)

Version: 2 Date: 5 March 2007

Author's response to reviews:

The ISRCTN number for this study is ISRCTN52943804.